Author's response to reviews

Title: Arthrofibrosis after TKA - Influence factors on the absolute flexion and gain in flexion after manipulation under anaesthesia

Authors:

Ingmar Ipach (IngmarIpach@gmx.de)
Falk Mittag (Falk.Mittag@med.uni-tuebingen.de)
Julia Lahrmann (julia.lahrmann@gmx.de)
Beate Kunze (Beate.Kunze@med.uni-tuebingen.de)
Torsten Kluba (Torsten.Kluba@med.uni-tuebingen.de)

Version: 6 Date: 10 August 2011

Author's response to reviews: see over
Dear editor,

Thank you for accepting our manuscript, in the following we performed the needed changes, so that the manuscript is now ready for publishing.

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name'

The list of authors is now exactly the same as in the submission system with the preferred style

Ingmar Ipach¹, Falk Mittag¹, Julia Lahrmann¹, Beate Kunze¹, Torsten Kluba¹

Authors' contributions: Please use the following format as an example, including using initials to refer to each author's contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript

All authors were totally involved in the manuscript.

Authors' contributions: II participated in the design of the study, statistical analysis and drafted the manuscript. FM participated in the design of the study and performed the statistical analysis. JL carried out the patient data and performed statistical analysis, BK performed the manipulation under anaesthesia and collected data. TK conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

Figures: Please provide a figure legend for figure 4 (and cite it as such) or place the table in the manuscript after the references and cite it as table 1. Please remove all vertical lines and shading.

We included table 1 in the manuscript and have removed all vertical lines and shading.

References: There are 14 references cited but only 13 in the reference list. Please cite 13 after 12 as this appears to be missing. Please also write details of reference 14 in the reference list.

There is no reference 14, therefore we placed 13 after 12 in the manuscript and have change the reference 14 into 13 in the reference list
Abstract: This should be identical in the manuscript and the submission system. Please amend this.

The abstract in the submission system is now the same as in the manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Line numbers: Please remove these.

Line numbers are removed

Affiliations: Each affiliation should be written in the following format: Department, Institute, (Street), City, (Postal code), Country.

Affiliations are now in the preferred style

Structure: Please change the heading "Introduction" in both the abstract and manuscript text to "Background". Please change "Materials and methods" to "Methods".

We performed the changes

Tracking: Please remove all coloured text and tracking.

All coloured texts are removed

CONFLICT OF INTEREST STATEMENT: please remove this.

Removed